Author's response to reviews

Title: Quality of Life as an outcome in Alzheimer's disease and other dementias - obstacles and goals

Authors:

Matthias W Riepe (matthias.riepe@uni-ulm.de)
Thomas Mittendorf (tm@ivbl.uni-hannover.de)
Hans Förstl (Hans.Foerstl@lrz.tu-muenchen.de)
Lutz Frölich (lutz.froelich@zi-mannheim.de)
Martin Haupt (m.haupt@alzheimer-praxis-duesseldorf.de)
Christoph Vauth (cv@ivbl.uni-hannover.de)
Reiner Leidl (leidl@bwl.lmu.de)
Matthias Graf von der Schulenburg (jms@ivbl.uni-hannover.de)

Version: 2 Date: 7 July 2009

Author's response to reviews:

Dear Editor:

We are grateful by the supportive and affirmative response of the reviewer. The reviewer’s remarks referred to unclear phrasing and we followed all of his suggestions for clarification (highlighted in yellow in the revised manuscript).

We would be grateful if the revised manuscript would be considered for publication in BMC Neurology.

Yours sincerely,
Matthias Riepe